Author's response to reviews

Title: Harnessing information and communication technologies to leverage scarce resources for cancer education, research and practice in developing countries.

Authors:
Valentine B Andela (info@cancer-africa.net)

Version: 3 Date: 21 December 2005

Author's response to reviews:

Attention:
Gemma Burns
Senior Editorial Production Assistant
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK
Tel: +44 (0) 20 73230323
Fax:+44 (0)20 7631 9923

Accepted manuscript - proofread and edited per instructions.